Author's response to reviews

Title: The discriminative power of patient experience surveys

Authors:

Dolf de Boer (d.deboer@nivel.nl)
Diana Delnoij (d.delnoij@centrumklantervaringzorg.nl)
Jany Rademakers (j.rademakers@nivel.nl)

Version: 4 Date: 17 November 2011

Author's response to reviews: see over
Dear BMC Editorial Production Team,

The following changes were requested. Our response is indicated by the usage of bold font:

- Authors' contributions: Please include the statement 'All authors read and approved the final manuscript'. **Response: statement is included**
- Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. **Response: done**
- Author list: Qualifications: Please remove all qualifications from the author list. **Response: done**
- Email address: please include email addresses for all the authors on the title page, using the following format:
  
  - AB: abcd@institution.ac.uk
  - EF: efgh@generic.co.uk
  - IJ: IJKL@corporation.com
  **Response: done**
- Highlighting/tracking: Please remove all highlighting and tracking from the manuscript. **Response: done**
- Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. **Response: done and several minor typographical errors were corrected.**

I hope the changes made are satisfactory,

Many thanks,

All the best,

Dolf de Boer